Contents lists available at ScienceDirect

Tourism Management

journal homepage: www.elsevier.com/locate/tourman

Tourism, terrorism and political violence in Tunisia: Evidence from Markovswitching models



Tourism Management

Charfeddine Lanouar*, Mohamed Goaied

Department of Finance and Economics, College of Business and Economics, Qatar University, P. O. Box 2713, Qatar

GRAPHICAL ABSTRACT



ARTICLE INFO

Keywords: Tourism Terrorism Jasmine revolution Markov shifts Persistence Spurious long memory Seasonality*JEL classification*: Z32 Z38 C52 C24

ABSTRACT

This study investigates the impact of terrorist attacks and political violence on the number of tourist arrivals and overnight stays in Tunisia. The dataset employed consists of monthly data that covers the period from January 2000 to September 2016, which includes several political and terrorist attacks in Tunisia and the region. Empirically, we investigate the true data generating process (DGP) of these two proxies of tourism activity by accounting for four statistical properties that characterize these series: (1) seasonality, (2) unit roots, (3) breaks, and (4) long memory behavior.

Our empirical findings show strong evidence of stationarity, five breaks in the tourist arrival time series and spurious long memory behavior. By estimating a 3-state Markov switching model consisting of the mean, trend, and variance, we find that the Tunisian Jasmine revolution and two recent terrorist attacks, one at the Bardo National Museum on March 18, 2015 and the other at the tourist resort at Port El Kantaoui, Sousse on June 26, 2015, played an important role in influencing the tourism activity of the country. Our empirical findings show also that local shocks have a more important impact than international shocks in influencing tourism activity. Interestingly, we find that the effects of terrorist shocks have a long duration compared to political violence shocks. Several security, marketing, and economic policies have been proposed and discussed in the paper.

1. Introduction

Throughout the world, tourism has become one of the industries most exposed to political unrest and terrorist attacks (Arana & Leon, 2008; Causevic & Lynch, 2013; Peter et al., 2014; Wolff & Larsen, 2014;

and; Avraham, 2016). In particular, since the beginning of the century, many democracies from developed countries have exhibited high levels of terrorist episodes. Starting with the September 11 Twin Towers attack in New York City in 2001, followed by the bombing events in Madrid in 2004 and London in 2005, and ending with the terrorist

https://doi.org/10.1016/j.tourman.2018.09.002



^{*} Corresponding author. E-mail addresses: lcharfeddine@qu.edu.qa (C. Lanouar), mgaied@qu.edu.qa (M. Goaied).

Received 5 August 2017; Received in revised form 2 September 2018; Accepted 3 September 2018 0261-5177/ © 2018 Elsevier Ltd. All rights reserved.

attacks in Brussels, Paris, and Berlin in 2016, terrorist incidents have created deep economic concern for the tourism industry. Recently, following the Arab Spring, an unprecedented wave of political violence and terrorist events have occurred in several MENA tourism-dependent economies, keeping tourists away from Egypt, Tunisia, and Turkey.

An important issue when proposing economic policies or designing a complete tourism strategy to restore tourism activity following political violence or terrorist attacks is the exact determination of the path time of these shocks. Indeed, the tourism industry is a sensitive sector and closely connected to other sectors (transport, trade, information and communications technology), so the social cost of such shocks can be damaging in terms of job losses and social exclusion if the recovery takes time. Therefore, the necessity to measure the time length of each shock is pertinent for policymakers and managers in the tourism industry. Furthermore, studies show that recovery is likely to be slow, as repeated incidents of violence tend to have long-lasting effects on tourist arrivals (IMF, 2017). To this end, it is important to determine whether these shocks have a transitory, persistent, or permanent impact. Estrada and Koutronas (2016) argue that "such severe shocks, concentrated in time, can cause major disruption on specific sectors of the economy." Based on these reports, it is important to focus on these shocks and study if political violence or terrorist attacks are transitory or persistent shocks, which will help to determine which type of policies should be recommended and implemented to recover from their negative effects.

Our choice of Tunisia as a case study of the impact of political violence and terrorist attacks on tourism activity is motivated by the growing place and important role that the tourism sector has in the Tunisian economy. Tunisia is a tourism-dependent economy, and the industry is considered one of the main drivers of the country's economic growth. For many years, the tourism sector has contributed to more than 7% of Tunisia's real GDP. In addition, this sector is one of the main sources of its foreign exchange. Furthermore, the tourism industry represents an important source of job creation, creating more than 400,000 jobs. Despite the importance of the tourism industry in the Tunisian economy, few studies have researched tourism in Tunisia. (see for instance Ouerfelli, 2008; Gasmi & Sassi, 2015; Jlassi, Fliti, & Chaibi, 2015; and Ben Aissa and Goaied, 2016).

The present study adds to the international tourism demand literature by examining and determining the true nature of the data-generating process (DGP) of two tourism activity proxies for the case of Tunisia (the number of tourist arrivals and the number of overnight stays). To this end, this paper explores several possible DGPs to model these two series. These alternatives include short memory models with breaks, processes characterized by unit roots, and models characterized by long-range dependence. Moreover, given the importance of the effective timing of tourism action, which can be drawn from modeling and determining the true DGP of the monthly time series, we investigate the short-run impact and the duration effects of two particular shocks in Tunisia: terrorist attacks and political violence shocks. To the best of our knowledge, this is the first study that attempts to examine the effects of those types of shocks on tourism activity using Tunisian data, recent econometric developments based on long-range dependence, and Markov switching models, as well as how to differentiate between these two processes.

The rest of this paper is organized as follows. Section 2 presents an overview of the literature review on terrorism and international tourism demand. Section 3 outlines the tourism sector in Tunisia and the different shocks that affect the series of the monthly number of tourist arrivals and nights. Section 4 presents the empirical methodology followed throughout the paper. Section 5 investigates the true DGP of the series of the number of tourist arrivals using several models. Section 6 uses Markov switching models with changes in the intercept, trend, and variance to model tourist arrivals and assess the robustness of our results to a second proxy of the tourism activity and a second technique of deseasonalization. Section 7 concludes the paper and

highlights the main policy implications.

2. Literature review

Terrorism and political violence have grown into a primary consideration for the tourism industry because these incidents seem to affect tourism activity more severely than all other types of shocks, such as natural catastrophes, financial crises, or health crises (Sonmez, 1998). Neumayear (2004) and Reisenger and Mavondo (2005) highlight the handicap effect of terrorism and political turmoil on the development of new destinations because consumer decisions show a preference for regions with more political stability and characterized by a peaceful social environment. Arana and Leon (2008) underlined the precarity of tourism following turbulences; both demand and supply can be sensitive to extreme events, such as terrorism or political violence. The empirical literature focusing on the impact of events (political violence, terrorist attacks, etc.) on international tourism demand can be classified into two groups.

2.1. Transitory or permanent shock on tourism demand

The first strand is based on determining whether the impact of such shocks on the tourism demand is transitory or permanent. Barros, Gil-Alana, and Wanke (2016) highlight that "a shock is known to have a temporary or short term effect if after a few number of periods the series return backs to its original performance level." For example, the impact of political violence and/or a terrorist attack shocks on the number of tourist arrivals vanishes quickly (after a few observations) and the demand recovers its primary realization. In contrast, there is evidence of a persistent or long-term impact, if the short-run effect of shocks converges toward a new trend in the achievement of demand.

Researchers have explored this possibility by applying unit root tests to tourism activity proxies (see Narayan, 2005, 2008; Lean & Smyth, 2009 among many others). For instance, using several unit root tests with structural breaks, Narayan (2005) shows that the 1987 political coups in Fiji had only transitory effects on tourist arrivals and tourist expenditures. Narayan (2008) investigates, in the context of panel data, the effect of shocks to international tourist arrivals in the case of Australia and shows these shocks have only a transitory impact. Lean and Smyth (2009) investigated the impact of shocks to international visitor arrivals from Malaysia's 10 major source markets. Using Lagrange Multiplier (LM) unit root tests with one and two structural breaks, the authors found evidence for stationarity of the tourist arrival time series and that the effects of shocks are transitory. These results are interpreted as evidence for long-term sustainability in Malaysia's tourist sector.

Recently, Raza and Jawaid (2013) investigated the impact of terrorism on the tourism sector in Pakistan during the period of 1980-2010. The authors show using the Johansen and Jeuselius and ARDL (autoregressive distributed lag) bound testing cointegration approaches that terrorist attacks have a significant negative impact on tourism in both the long and short runs. More recently, Liu and Pratt (2017) measured the relationship between terrorism and tourism based on international demand models using panel data for 95 countries. Their findings indicate that globally, tourism is resilient to terrorism, with a limited impact in the short run. Interestingly, most of the literature on the terrorism-tourism nexus is based on annual time series or panel data. Agiomirgianakis, Serenis, and Tsounis (2017) argue that government actions and policies on tourism activities require being prepared for a quite short-term objective. Indeed, he stresses that political instability and terrorist upheavals, among other short internal or external factors, are often unpredictable for terms longer than 6-9 months. Furthermore, this timing dependence of policy decision makers (governmental and private) in the tourism industry is a problem the literature often fails to address. Our objective is to fill this gap fairly.

2.2. Persistence property of shocks on tourism demand

The second and more recent strand has focused on measuring the persistence property of shocks on international tourism demand. Instead of using unit root tests, most of the recent studies have investigated whether tourism activity series are characterized by a fractional integration parameter between zero and one.¹ Most of these studies have employed monthly or quarterly data of foreign tourist arrivals and/or overnight stays (Gil-Alana, 2005; Cunado, Gil-Alana, & de Gracia, 2008a; 2008b; Chu, 2008; and Gil-Alana, De Gracia, & CuÑado, 2004, 2008, 2014). Using quarterly data on the total number of arrivals in Kenva, Gil-Alana, Mudida, and de Gracia (2014) reveal that the series are fractionally integrated with an order above zero and below 1. Hence, they anticipate that shocks persistently fade slowly. Gil-Alana, Mervar, and Payne (2015) examine the presence of a long memory behavior in monthly foreign tourist arrivals and overnight stays in the Adriatic region of Croatia. Using parametric and semiparametric fractional integration approaches, their findings reveal evidence for the persistence of shocks for most of the Croatian coastal counties.

However, several empirical studies have shown that it is important to be careful when examining and investigating the long memory property as this property can be spuriously created by structural changes that affect the short memory process (see Charfeddine & Guégan, 2012). This evidence of confusion between true and spurious long memory behavior was first revealed by Diebold and Inoue (2001), svirta (1999), and Jasiak (2001), Granger and Terosvirta (1999), and Granger and Hyung (2004). In the tourism literature, many researchers have examined the presence of long-range behavior for many countries by exploring several tourism activity proxies (see Gil-Arana & Leon, 2008 among many others). Moreover, they have examined possible discrimination between a case where the long-range dependence behavior is true and the case of spurious behavior, which is created by neglecting breaks inside time series. Ohanissian, Russell, and Tsay (2008) argue that observing a long-range dependence behavior and discriminating between true and spurious behaviors are a difficult issue. Gil Arana and Leon (2008) and Mayral (2012) highlighted that the concepts of fractional integration and structural breaks are closely related issues. To account for both aspects, Gil-Alana et al. (2014) test the degree of persistence of the series with the presence of structural breaks, where the number of breaks and the dates are endogenously determined by the model.

2.3. Persistence of shocks and seasonality issues

Barros et al. (2016) stress the importance of correctly handling seasonality when measuring the persistence of shocks. Indeed, Song and Li (2008) argue seasonality is a striking feature of monthly or quarterly tourism data that needs to be considered when modeling and forecasting tourism demand. Gil-Alana, Cunado, and Perez de Gracia (2008) suggest that modeling the element of seasonality in a chronological series continues to be a controversial issue. Authors have considered deterministic and stochastic seasonality (seasonal unit roots and seasonally fractionally integrated models) when forecasting the number of tourists travelling to the Canaries Islands. Gil-Alana, Mervar, and Payne (2016) examined the impact of shocks on tourism in Croatia. They found that monthly foreign tourist arrivals and overnight stays contain seasonal unit roots, with the seasonally differenced components of the series allowing for transitory shocks to vanish relatively quickly. Ouerfelli (2008) applied a cointegration analysis and error correction models to study the profile of the European tourism market in Tunisia, based on quarterly tourist arrivals data from 1981:1 to 2004:4. Seasonal fluctuations in European tourism demand in Tunisia have been assessed using HEGY tests (Hylleberg, Engle, Granger, & Yoo, 1990) to test the presence of seasonal unit roots. The results show that these series exhibit non-stationary stochastic seasonality.

In the theoretical literature, the question of which method of deseasonalization should be used has been examined by many researchers, including Hood, Ashley, and Findley (2000), Franses, Paap, and Fok (2005) and many others. Hood et al. (2000) compare the ×12-ARIMA technique to the TRAMO/SEATS method. Their results show that overall these two techniques perform similarly for simulated series using 12 years of data. Moreover, TRAMO/SEATS perform better if the series is subject to large irregularities. The only case where $\times 12$ -ARIMA performs better is if the data sample is small. This result of similarity in terms of performance is also found by Franses et al. (2005, pp. 2005–2030). These researchers compare the performance of the TRAMO/SEATS, ×12-ARIMA, and Dianties methods using several DGPs. They find that TRAMO/SEATS and $\times 12$ -ARIMA perform well compared to the Dianties technique and whatever the DGP used. Moreover, their results show that the presence of additive outliers do not have a serious effect on the performance of these two methods. The authors recommend using one of these two techniques when there is no indication of the type of seasonality. In this paper, as our data spans more than 12 years, we propose the use of the TRAMO/SEATS technique; then, we propose using $\times 12$ -ARIMA as a robustness check.

3. The tourism sector in Tunisia

Historically and before the 2011 revolution, Tunisia was an attractive destination for many European countries, thanks to its affordability and close proximity to Europe. After its independence, the strategy of Tunisian tourism was exclusively targeted on the European Market in search for a nearby, safe, and affordable sun and sea destination. Presenting Tunisia as a safe and low-cost destination was a plan that succeeded in making it a player in the international tourism market for the last three decades. Carboni, Perelli, and Sistu (2014) mentioned that this strategy reinforced the dependence of the Tunisian tourism industry on European tour operators, which could undermine the Tunisian economy because of external shocks that may have an impact on employment and the external equilibrium. In the same vein, Jlassi et al. (2015) highlighted that the negative effects of the above-mentioned strategy rely in exposing the destination country Tunisia) to the shocks of the origin countries, such as the effect of the European crisis on tourism in Tunisia.

The tourism activity in Tunisia in the last two decades has shown high fluctuation, particularly during and after the periods of political instability or terrorists attacks. For instance, since the Jasmine revolution of January 2011, there has been a sharp fall in most of the Tunisian tourism indicators. For instance, between 2010 and 2015, the number of tourist arrivals, tourist overnight stays, and tourism receipts fell by 31.5%, 54.5%, and 33.2%, respectively (Authors' calculation based on data provided by the national institute of statistics of Tunisia.). Moreover, since 2010, the Tunisian economy has been subject to several local and international shocks, including several national, regional, and international shocks, for example, terrorist attacks against Tunisian national polices and several tourist sites, the Egyptian and Libyan revolutions, and several other shocks that have threatened the overall activity of the sector. This sharp fall in most of the tourism indicators during the last few years, combined with several regional and local events (terrorist attacks), have motivated us to examine the effects of these recent political and terrorist attacks on tourism activity. This

¹ Shocks are considered as persistent, in the sense that they have long-lasting effects, if the estimated fractional degree of differentiation of the series, d, lies above zero and below one. In the case when d = 1, the effect of shocks are permanent. If the fractional long memory parameter d is not significantly different from zero, then the impact of shocks on the tourism activity will vanish and die after a few observations (a few months in our case). For more details, see Charfeddine and Guégan (2012), who summarize the difference between unit root tests, unit root tests with breaks, and the fractional integration procedure.



Fig. 1. Trends in Tunisian tourism indicators in Tunisia. 2000–2015. Source: Authors calculations and Tunisian National Institute of Statistics.

examination will help us to propose some economic, security and marketing policies that may help to restore Tunisian tourism activity.

Fig. 1 shows the trend in some tourism activity indicators in Tunisia for the period from 2000 to 2015. This figure sheds light on the question of whether terrorist activities have an impact on the tourism industry. The slowdown of tourism receipts and tourism GDP from 2002 to 2003 is explained by the aftermath of September 11 and the effect of the June 2002 terrorist attack in Djerba. In the immediate aftermath of more recent events, including the 2011 revolution, the drop in tourism activities exacerbated structural issues, including a lack of diversified offerings and the tourism industry's over-dependence on European markets. In 2015, another drop in tourism activities was observed following two terrorist incidents: in March at the Bardo Museum in Tunis and in June at the Resort in Sousse. During 2015, tourist arrival fell to 5.5 million, the lowest level in decades, and tourism revenues fell 35% to USD1.5 billion. According to the ministry of tourism, the fall in tourism revenues contributed to a slowdown in economic growth to 0.8% from 2.3% a year earlier. It is interesting to note, from Fig. 1, that external shocks (such as the September 11 terrorist attack or the financial crises in 2008) had a moderate impact on Tunisia's tourism activity in comparison with internal events (such as the political instability after the Jasmine Revolution in 2011 or the terrorist attacks in 2015). This underscores the business strategy of anticipating disturbances in the tourist landscape because of terrorist attacks, which may adversely affect tourism activity and have negative social consequences in terms of job losses and poverty acceleration. Unfortunately, graphic analyses do not distinguish between the impact of political violence and terrorist attacks, both of which seem to have severe consequences for tourism activity.

4. Data and preliminary analysis

The aim of this section is to determine the true data generating process of the tourist arrival series. To this end, several statistical properties of the tourist arrival series will be examined in the next subsections. These properties include seasonality, trends, unit roots, structural breaks, long memory, and spurious long memory.

In subsection 4.1, we will investigate all these properties from a descriptive point of view. We will do this by analyzing the trajectory, autocorrelation function (ACF) and spectral density of the number of tourist arrivals. Then, in subsection 4.2, we will employ several testing procedures to determine the true data generating process of the number of tourist arrivals. In particular, we will test for three important properties: (1) unit root with structural breaks, (2) long memory, and, finally, and (3) true versus spurious long memory.

4.1. Data description

The dataset used in this study consists of monthly data for the number of international tourist arrivals in Tunisia from January 2000 to September 2016, which were collected from the Tunisian National Institute of Statistics from monthly publications: "Bulletin mensuel de statistique". Annual data regarding tourism activities, such as tourism receipts, investments, occupation rates, and nights, can be found at http://dataportal.ins.tn/. Fig. 2 displays the trajectory, autocorrelation function, and spectral density of the seasonally unadjusted (Panel A) and seasonally adjusted (Panel B) series.

From Panel A, and specifically from the trajectory of the seasonally unadjusted monthly number of tourist arrivals, we observe that these series are characterized by three main statistical properties: (1) seasonality, (2) structural changes, and (3) trends.

The first property of seasonality, defined by the presence of a regular fluctuation in the number of tourist arrivals, is observed in the trajectory in months June, July, and August. During these months, the number of tourist arrivals reached its highest levels, a result that is expected as these three months correspond to the periods of the tourist summer season and summer vacation of almost all the European countries and Tunisia neighbor countries (Algeria and Libya).

The second statistical property, the presence of structural changes, is also visually observed in the trajectory of the number of tourist arrivals, in which at least two sudden changes are observed. The first break coincides with approximately the end of the year 2010 and the second one coincides with mid-year 2015. From a political view point, these two breaks seem to be associated with the Tunisian Jasmine revolution of January 2011 and the recent two terrorist attacks in March and June 2015 against the Bardo National Museum and the hotel of Port of Kantaoui, Sousse. For instance, as shown in the introduction, the Jasmine revolution and the last two terrorist attacks in Tunisia have severely impacted the tourist sector, particularly the number of tourist arrivals.

Finally, the last property that characterizes the number of tourist arrivals is the trend property, which is observed in the trajectory of the seasonally unadjusted tourist arrival series. For instance, the trajectory of this later series shows clear evidence for the presence of an upward trend during all periods of study. In particular, the trajectory shows evidence of the presence of at least three upward trends that seem to characterize the tourist arrival series. The first trend characterizes the period from January 2000 to January 2011, the second characterizes the period from January 2011 to March 2015; and the third characterizes the rest of the period of study.

Moreover, in addition to all three properties, the ACF and spectral density of the tourist arrival series show that these data may be



Fig. 2. Trajectories, Autocorrelation function (AFC) and spectral density of the monthly number of adjusted and unadjusted tourist arrivals series.

characterized by of the presence of a long memory component. Evidence for the presence of this property is observed and detected by visual analysis of the autocorrelation function (ACF), which decays slowly to zero, and the spectral density, which shows peaks at different frequencies.

In addition to the trajectory, ACF and spectral density graphics reported in Panel A for the original series, we report in Panel B these same graphics but for the seasonally adjusted number of tourist arrivals. To remove the seasonal component, we use the TRAMO/SEATS procedure. The use of this method is motivated by its superiority compared to other approaches, including the well-known X-12-ARIMA technique (Franses et al., 2005, pp. 2005-2030). Moreover, we use X-12-ARIMA as a second technique of de-seasonality to increase the robustness of our results. The use of the second method is motivated by the empirical findings of Barros et al. (2016), who found that if seasonality is not accurately managed, then the persistence of shocks will not be appropriately resolved. Our empirical results show that after removing seasonality from the original series, the new trajectory shows the same evidence for the presence of both structural breaks and trends. Overall, there are no strong changes compared to the results from the original series. In particular, the trajectory shows clear evidence that the number of tourist arrivals was volatile between January 2000 and mid-2003, followed by a continuous increase until January 2011. Between January 2011 and July-August 2011, there was a significant and intense decline in the number of monthly tourist arrivals, which can be explained by the effects of the Jasmine revolution, which lasted from December 2010 to January 2011. Finally, we observe an increase in the number of tourist arrivals until mid-2015, at which point a new drop is observed until the end of the period of our study. The economic explanation of the break dates are as discussed above.

To account for all these statistical properties, we propose in the next section a statistical procedure that allows us to ascertain the true DGP of the tourist arrival series. This testing procedure will make use of several tests' statistics, including unit root tests with and without breaks, and tests for true versus spurious long memory behavior.

4.2. Data generating process (DGP) of the tourist arrivals

To build and design appropriate strategies and/or policies that allow Tunisian policymakers to restore the tourism activity of the country in an efficient way, it is of primary importance to determine the true DGP of the tourist arrivals time series. For instance, the extent of the impact of shocks in term of duration and magnitude differs significantly depending on the true DGP of the tourist arrival series. Therefore, it is very important to test the two extreme cases of transitory versus permanent and the two cases of long memory versus short memory with structural breaks (Diebold & Inoue, 2001; Huang and Geweke & Porter-Hudak, 1983; and; Charfeddine & Guégan, 2012).

In this study, we propose an empirical strategy defined in four steps to determine the true DGP of the series under study:

- 1. First, we begin our analysis by testing for whether the impact of shocks on the number of tourist arrivals is transitory against the case where shocks have a permanent effect. To investigate these two possibilities, we employ both unit root tests with and without structural breaks. More precisely, we use four traditional unit roots tests as developed by Ng-Perron (1995) and their counterparts that allow for multiple unknown breaks up to five breaks under both the null and the alternative hypothesis (see Charfeddine & Guégan, 2011).
- 2. Then, we use several long memory estimation methods and test statistics to validate the evidence of long memory observed in the ACF and spectral density. These methods and tests include the modified R/S test of Lo (1991), the rescaled variance (V/S) test of Giraitis, Kokoszka, Leipusc, and Teyssiere (2003), the Geweke and Porter–Hudak (Geweke & Porter-Hudak, 1983) test, the Gaussian semi-parametric (GSP) test of Robinson and Henry (1999), the Local Whittle (LW) test of Robinson (1995), the exact local whittle (ELW) test of Shimotsu and Phillips (2005), and the 2-step exact local Whittle (2SELW) test of Shimotsu (2006b). All these methods and tests are used to explore the long memory behavior observed in the ACF and spectral density.
- 3. Third, we infer and test, by using the sample splitting test of

Shimotsu (2006a), the two behaviors of short memory with structural breaks (Charfeddine & Guégan, 2011) and long-range dependence. The outcomes of this step will provide us with the true DGP of the number of tourist arrivals.

4. Finally, based on the results obtained in 3, a flexible autoregressive Markov switching model or a long memory model will be employed to better model the number of tourist arrivals. However, based on the results of section 4, a Markov switching model where the intercept, trend, and variance all switch between regimes seems to be more appropriate for modeling the number of tourist arrivals in Tunisia.

4.2.1. Unit root tests with breaks

One of the merits of applying unit root tests to the number of tourist arrivals is the ability to distinguish between the two extreme behaviors of transitory and permanent effects of shocks. Econometrically, evidence of stationarity for the tourist arrival series indicates that the effect of shocks will be transitory, i.e., the impact will vanish after a few observations (a few months in our case). In contrast, a non-rejection of the null hypothesis of unit root means that the impact of shocks is permanent and strong policies should be implemented to restore tourism activity.

To examine the transitory versus permanent effects of shocks on tourism activity, proxied by the number of tourist arrivals, we use both the traditional unit root tests of Ng-Peter, Poulston, and Losekoot (2014) and the unit root tests with structural breaks of Carrion-i-Silvestre, Kim, and Perron (2009). The main motivation behind the use of unit root tests with structural breaks lies in the fact that ignoring the presence of breaks in the tourist arrival series, when testing for the presence of unit root, can introduce a misleading and incorrect interpretation of the results. For instance, traditional unit root tests are known to suffer from low power when the time series under study are subject to structural breaks. Moreover, the unit root tests of Carrion-i-Silvestre et al. (2009) have the advantage of considering the case of unknown multiple structural changes up to five breaks under both the null and alternative hypotheses.² These advantages will help us better understand whether shocks on the tourist arrival series will be transitory or permanent.

Four traditional unit root tests of Ng-Perron (1995) and their counterpart unit root tests with structural breaks, as developed by Carrion-i-Silvestre et al. (2009), will be used in this study. These latter tests include the $MZ_{\alpha}^{CLS}(\lambda)$, $MSB^{GLS}(\lambda)$, $MZ_t^{CLS}(\lambda)$, and $MP_T^{GLS}(\lambda)$ tests, where λ is the break point parameter. The limiting distribution and critical values of all these tests are obtained by simulation (a detailed description of all these tests is presented in the supplementary document).

The results of applying both types of unit root tests on the time series of the number of tourist arrivals are reported in Table 1. The results of the Ng-Perron (1995) unit root tests reported in Panel A show mixed evidence regarding the stationarity hypothesis. For instance, whereas MZ_{α}^{GLS} and MSB^{GLS} show strong evidence of stationary using the intercept with/without trend specification, the MZ_t^{GLS} and MP_T^{GLS} statistics do not reject the hypothesis of unit root when allowing for an intercept and trend.

The empirical results of the unit root tests of Carrion-i-Silvestre et al. (2009), reported in Panel B of Table 1, show that the hypothesis of

Table 1

Unit root with and without structural changes for logarithm of	arrivals.
--	-----------

Panel A:	Unit root	without	structural	breaks	

	With intercep	ot	With intercept	and trend
	Statistics	C·V	Statistics	C·V
MZ_{α}^{GLS} MZ_{t}^{GLS} MSB^{GLS} MP_{T}^{GLS}	-7.149 -1.809 0.253 3.726	-8.100 -1.980 0.233 3.170	- 16.931 - 2.9074 0.1717 5.3954	- 17.300 - 2.9100 0.1680 5.4800

Panel B: Unit root with structural breaks

	Crash mod	el (shifts in l	evel only)	Shifts in level and slope of the tree			
	Statistics	C·V	Breaks dates	Statistics	C·V	Breaks dates	
$\begin{split} & MZ^{GLS}_{\alpha}(\lambda) \\ & MZ^{GLS}_{t}(\lambda) \\ & MSB^{GLS}(\lambda) \\ & MP^{GLS}_{T}(\lambda) \end{split}$	- 47.968 0.1020 - 4.8943 6.2293	- 38.819 0.1149 - 4.3865 7.9071	2001- M09 2003- M01 2011- M01 2011- M09 2016- M06	- 57.996 0.0928 - 5.3848 5.1331	- 38.802 0.1139 - 4.3971 7.8227	2001- M09 2011- M01 2011- M09 2014- M06 2015- M06	

stationarity with structural breaks, the I(0) process, cannot be rejected regardless of the unit root tests used and specification employed (intercept with or without trend). This evidence of stationarity indicates that the effect of shocks on tourist arrivals is transitory with evidence of structural breaks.

In particular, the results show that the series of the number of tourist arrivals is characterized by the presence of five breaks. Interestingly, these break dates correspond to several political events and terrorist attacks. In particular, we found evidence of a significant effect for the September 11 (2001) terrorist attack in New York City, the January 2011 Jasmine revolution in Tunisia, and the recent terrorist attacks at the Bardo National Museum and the Sousse Port Kantanoui hotels between March 2015 and May 2015. This result confirms our preliminary analysis based on the trajectory of the number of tourist arrivals.

4.2.2. The long memory property

i Definition of long memory

As mentioned above, one of the properties to be explored is the property of long memory. In the econometric literature, there are two possible definitions of long memory. The first is a time domain definition. Following this definition, a time series (Y_t) , t = 1, ..., T is said to display a long memory behavior if its autocorrelation function decays slowly (hyperbolically) towards zero. In a mathematical presentation, the autocovariance function of Y_t will be

$$\gamma_k \approx c(k) \, k^{2d-1},\tag{1}$$

where *d* is the fractional long memory parameter. If 0 < d < 1, the process is a long memory process, if $0 < d < \frac{1}{2}$, the long memory process is stationary, and if $\frac{1}{2} < d < 1$, the long memory process is non-stationary but mean-reverting. c(k) is a slowly varying function at infinity.

The second definition of long memory is a frequency domain definition. In this case, the series (Y_t) exhibits a long memory if the spectral density explodes at frequency near zero. The frequency domain definition of long memory is given by

² Compared to the several unit root tests with structural breaks, such as the tests of Zivot and Andrews (1992), Lee and Strazicich (2003), and Lumsdaine and Papell (1997), the unit root tests with structural breaks proposed by Carrion-i-Silvestre et al. (2009a) have the advantage of considering the case of multiple breaks where the dates of these breaks are unknown (endogenously determined). The maximum number of breaks allowed under these tests is five breaks. Moreover, compared to the majority of previous tests in the empirical literature, except those of Lee and Strazicich (2003), these tests assume the presence of breaks under both the null and alternative hypotheses.

Table 2

Estimation	of th	ne	fractional	10	ng	memory	parameter	for	the	tourist	arrivals	time	series.
Dottinteron	~ ~	· •	machioman			momory	paramotor				and the terro	c	001100.

	Non-parametric tests		m	Parametric tests					
q	Modified R/S	V/S	_	GPH	GSP	LW	ELW	2S-ELW	
1 5 10 20	2.8893*** 1.9877** 1.4512 1.2708	0.9210*** 0.4255*** 0.2401** 0.1594	$T^{0.5}$ $T^{0.6}$ $T^{0.7}$ $T^{0.8}$	0.5823*** 0.7898*** 0.7362*** 0.7485***	0.5507*** 0.6338*** 0.5720*** 0.5701***	0.2598*** 0.1829** 0.1354* 0.1082	0.9416*** 0.9298*** 0.9989*** 0.9928***	0.5845*** 0.6443*** 0.9320*** 0.6067***	

The critical values of the Modified R/S statistic see Lo (1991) and for the critical values of the V/S statistic see Giraitis et al. (2003).

The critical values of the GPH, GSP, LW, ELW and 2S-ELW statistics are those of the standard normal distribution, 1.645, 1.96 and 2.58 for the 10%, 5% and 1% respectively.

*,**,*** stand for significance at the 10%, 5% and 1% level of significance respectively.

$$f(\omega) \approx \infty \text{ as } \omega \text{ tends to zero}$$
 (2)

ii Testing for long memory

To test for long memory behavior in the tourist arrival series, we propose the use of two non-parametric tests statistics of long memory, namely the modified R/S and V/S statistics and five semi-parametric estimation methods of the fractional long memory parameter *d*. These later methods include the GPH, GSP, LW, ELW, and 2SELW estimation techniques. The estimation results are also reported in Table 2, where different values of truncation lag parameter are used. For the two nonparametric statistics, we use q = 1, 5, 10, and 20. For the parametric tests statistics, we use $m = T^k$, where k = 0.5, 0.6, 0.7, 0.8 and *T* is the number of observations.

The results show that the number of tourist arrivals is characterized by the presence of long memory behavior. For instance, five out of the eight calculated statistics of the modified R/S and V/S tests show evidence for the rejection of the null hypothesis of short memory against the alternative of long-range dependence. In contrast, 18 out of the 20 estimated fractional long memory parameters, using the parametric tests, are significantly different from zero.

In terms of economic significance and implications, evidence of the presence of long-range dependence, if it is a true behavior, indicates that shocks affecting the number of tourist arrivals have long-lasting effects and their effects do not disappear rapidly. However, it is worth noting that before continuing with the discussion of the economic interpretation and the policy implications of this latter result, it is important to examine whether this long memory behavior is true and not spuriously created by the presence of structural breaks (see Diebold & Inoue, 2001; Huang and Geweke & Porter-Hudak, 1983).

Indeed, we need to be cautious, as many empirical studies have shown that the existence of breaks can be attributed to the existence of a long memory behavior in the time series and that these breaks are spurious (Beran, 1994). In contrast, several other studies have shown that long-range dependence behavior can be created by the presence of breaks inside these series (Diebold & Inoue, 2001; Huang and Geweke & Porter-Hudak, 1983; and; Charfeddine & Guégan, 2012; Walther, Klein, Thu, & Piontek, 2017). To investigate this issue of true versus spurious long memory, we propose the use of the splitting sample test of Shimotsu (2006a), which is designed to distinguish between short memory with breaks (spurious long range dependence) and true longrange dependence behavior. The results of this test will help us to be clear cut in term of the distinction between these two different behaviors and in determining the true data generating process of the number of tourist arrivals.

4.2.3. Shimotsu (2006a) test of true versus long-range dependence

The Shimotsu (2006a) test consists of splitting the tourist arrivals time series into b subsamples and in comparing the estimated long memory parameter d of the entire sample to the estimated fractional long memory parameter of each subsample. The idea of this test is based on the following assumption: If the observed long memory behavior is a true behavior, then the estimated fractional long memory parameter of d for the full sample should not be significantly different from those estimated from the subsamples.

The null hypothesis of the true long memory I(d) is as follows:

$$H_0$$
: True long memory(H_0 : $d = d^{(1)} = \dots = d^{(b)}$) (3)

The alternative hypothesis corresponds to the following:

$$H_1$$
:Short memory affected by breaks (4)

where $d^{(i)}$ for $i \in \{1, 2, ..., b\}$ is the true value of the fractional long memory parameter from the ith subsample and d is the true value for the full sample.

The test statistic proposed by Shimotsu (2006a) is called the W_c test, which has a chi-squared limiting distribution with b - 1 degrees of freedom (see the Appendix for a full description of the test).

The results of the splitting sample test (W_c) of Shimotsu (2006a) are reported in Table 3. We use the values of b = 5 and b = 6. These choices are based on the number of breaks selected in the previous section: five breaks (b = 6). We also consider the case where the number of breaks is 4 (b = 5). The results of the W_c test show that the hypothesis of true long memory behavior is rejected whatever the value of the parameter of truncation m and for both values of b. We confirm the results of the previous section, which indicate that the DGP for tourist arrivals is a short memory process that has been subject to several structural breaks.

5. A markov switching model for the number of tourist arrivals

Once the evidence for stationarity with structural breaks for tourist arrivals has been approved and well established, the next step consists of modeling the tourist arrivals time series by using models that can account for both short memory and structural breaks. In econometric theory, several models have been proposed to account for these two behaviors, which include the threshold autoregressive models (TAR) of Tong (1990) and their different extensions (Terasvirta, 1994; and; Terasvirta & Yang, 2014), the structural change models (Quandt, 1958; and Bai & Perron, 1998, 2003), and the Markov switching model of Hamilton (1989). Several other models exist that take into consideration the two behaviors of short memory and breaks. These models

Table 3

Shimotsu (2006a) test for true versus spurious long memory for the number of tourist Arrivals.

Shimotsu (200	Shimotsu (2006a) Splitting sample test (W _c)						
	m	Logarithm of Number of tourist Arrivals					
<i>b</i> = 5	20	10.322**					
	40	15.083***					
b = 6	20	15.173***					
	40	18.519***					

include the mean-plus-noise model of Diebold and Inoue (2001), the sign model of Terasivirta and Geweke and Porter-Hudak (1983) and the STOPBREAK model of Engel and Smith (1999), among many others.

In this study, we propose the use of a general form of the Markov switching autoregressive model where all the parameters, except the autoregressive order, are allowed to switch between regimes. We choose the Markov switching autoregressive model because it encompasses most of the models presented above as a special case (see, for instance, Krolzig, 1997; Carrasco, 2002; and Charfeddine & Guégan, 2008).

5.1. The Markov switching autoregressive model

Because the hypothesis of true long memory was rejected using Shimotsu (2006a) test against the alternative of short memory with breaks, the next step consists of using models that take into consideration short memory and changes in regimes. We propose the use of the Markov switching Autoregressive (MS(k) - AR(p)) model with changes in both the mean and variance, where *k* represents the number of regimes and *p* is the autoregressive order.

The general form of the MS(k) - AR(p) model is given by the following:

$$y_t = \mu_{s_t} + \delta_{s_t} \oplus trend + \sum_{i=1}^p \varphi_i y_{t-i} + \sigma_{s_t} u_t$$
(5)

where s_t is a dummy variable that takes values of 1, 2 and 3. It is governed by a first-order Markov chain that is given by the following:

$$S_{t} = \begin{cases} 1 \text{ with probability } p_{11} \\ 2 & \text{with probability } p_{22} \\ 3 & \text{with probability } p_{33} \end{cases}$$
(6)

where

$$P = \begin{pmatrix} p_{11} & p_{12} & p_{13} \\ p_{12} & p_{22} & p_{23} \\ p_{31} & p_{32} & p_{33} \end{pmatrix} \text{ and } \sum_{j=1}^{3} p_{ij} = 1$$
(7)

Under these notations, the intercept and slope coefficients can be written as follows:

$$\mu_{S_t} = \mu_1 \, S_{1t} + \mu_2 \, S_{2t} + \mu_3 \, S_{3t} \tag{8}$$

and

$$\sigma_{s_t}^2 = \sigma_{s_t}^2 S_{1t} + \sigma_{s_t}^2 S_{2t} + \sigma_{s_t}^2 S_{3t}$$
⁽⁹⁾

where $S_{kt} = 1$ if $S_t = k$ and $S_{kt} = 0$; otherwise, k = 1,2,3.

To select the most appropriate specification of the MS(k) - AR(p), we proceed in the following way:

- 1. First, the automatic model selection technique, which is implemented by using the generic-to-specific (*Gets*) approach, is used to select the most appropriate autoregressive lag order. The main advantages are that it cannot be outperformed even by an expert in econometrics and it is not time consuming. The automatic implementation of the *Gets* technique requires only the selection of the initial model presentation, known in the literature as the general unrestricted model, and then the computer will select the more appropriate presentation by checking the validity of each model reduction using a large number of residual-based test statistics.
- 2. The selected AR(p) specification will be used as the starting point for estimation procedure of the Markov switching autoregressive model, MS(k) - AR(p). Several specifications of the MS(k) - AR(p)model will be examined by allowing the intercept, trend, autoregressive order coefficients and variance to depend on the state of the economy, S_t . Note that in this study, we examined several variance specifications (fixed variance, variance switching, GARCH switching, mean-variance component). However, as we use a non-

financial time series, we expect that the GARCH switching and mean-variance component will not fit the variance structure of our data.

- 3. To select between the linear AR(p) and the Markov switching autoregressive model MS(k) AR(p) and between different MS(k) AR(p) models, we use the log-likelihood ratio (LR) test. An important remark is that the LR test has a non-usual limiting distribution under the null hypothesis. To solve this problem, Davies (1977, 1987) proposed a bound test approach that allows us to test the linear versus Markov switching specification. Hansen (1992) and Garcia (1998) also developed other tests statistics and approaches (see Charfeddine & Guégan, 2011 and 2012).
- 4. Finally, after selecting the adequate Markov switching autoregressive model, to assess the quality of the fitted model, we use the regime classification measure (RCM) of Baele (2005):

The RCM is given for K > 0 states by the following:

$$RCM(K) = 100* \left(1 - \frac{K}{K-1} \frac{1}{T} \sum_{t=1}^{T} \sum_{h=1}^{K} \left(p_{i,t} - \frac{1}{K} \right)^2 \right)$$
(10)

where $p_{i,t} = \Pr[S_t = i|I_T]$ is the probability of being in regime *i* at time t. The *RCM* measure lies between 0 and 100. The decision rule when using this measure is that if the true DGP is a Markov switching model, then the diagonal elements of the transition probabilities matrix will be close to the unit, which makes the *RCM* measure close to zero. In contrast, if the DGP is not a Markov switching model, then the *RCM* will be close to 100.

A second tool used to assess the quality of the selected Markov switching model is to analyze and compare its residuals properties, in terms of the absence of autocorrelation and homoscedasticity, to the linear and MS(2) - AR(p)) models.

An important issue that arises when using the Markov switching models is that the maximum log-likelihood technique can encounter several local optimums, rather than a unique global optimum (see Hamilton, 1989, 1990). To solve this problem, Hamilton (1989) suggests using several starting values. In this study, we use more than 1000 starting values for each estimated specification to guarantee that the maximum obtained is a global one.

5.2. Markov switching model selection and interpretations

5.2.1. Model selection

Estimating the Markov switching model involves selecting the lag parameter p and the parameters that switch between regimes and determining the number of regimes. The value of autoregressive lag p is selected using the automatic model selection technique implemented via the *Gets* approach, which shows that a first-order autoregressive model is enough for this time series to remove the presence of autocorrelation. The results also show that by estimating different specifications of the MS(k) - AR(1) model, there is strong evidence for switching in the intercept, trend, and variance. This evidence of intercept and trend parameter switching confirms the results previously obtained by the Charfeddine and Guégan (2011) unit root tests with structural changes.

The results of the LR test for testing between linear versus MS(2) - AR(1) and between the models MS(2) - AR(1) and MS(3) - AR(1), reported in Table 4, support the alternative hypothesis of MS(2) - AR(1) against the null hypothesis of a linear model, and the alternative hypothesis of MS(3) - AR(1) against the null hypothesis of MS(2) - AR(1).³ This result of the superiority of the MS(3) - AR(1)

³ Testing between different numbers of regimes in the Markov switching model remains a very complicated task for several problems, such as the existence of nuisance parameter under the null hypothesis, identically zero scores and the existence of several local maxima. Davies (1977, 1987) proposed an

Table 4

Estimation of the linear, Markov switching with 1 and 2 breaks.

	Linear Model		MS(2)-AR(1)		MS(3)-AR(1)		
	Coef.	t-stat	Coef.	p-value	Coef.	p-value	
Models Param	eters estimation	ons					
μ_1	2.4414	0.000	0.8068	0.127	10.737	0.000	
μ_2	-	-	1.1295	0.004	10.628	0.000	
μ_3	-	_	-	_	10.118	0.000	
φ_1	0.8124	0.000	0.9141	0.000	0.1695	0.000	
@trend1	9.724e-05	0.440	0.0022	0363	0.0022	0.000	
@trend2	-	-	-4.67e-05	0.598	0.0017	0.000	
@trend3	-	-	-	-	0.0030	0.000	
σ_1	0.0924	0.000	0.3524	0.000	0.0410	0.000	
σ_2	-	-	0.0607	0.000	0.0714	0.000	
σ_3	-	-	-	-	0.0762	0.000	
p_{11}	-	-	0.9617	0.000	0.9837	0.000	
p_{12}	-	-	-	-	0.0130	0.438	
<i>p</i> ₂₁	-	-	-	-	0.0084	0.755	
<i>p</i> ₂₂	-	-	0.6712	0.000	0.9667	0.000	
p ₃₂	-	-	-	-	0.0130	0.459	
P ₃₃	-	-	-	-	0.9570	0.000	
Log. L	181.958		202.499		254.278		
LL ratio (LR)			41.082		103.558		
RCM	_		8.355		1.840		
AIC	-1.8831		-2.3757		-2.5320)	
SC	-1.8145		-2.2384		-2.2748	3	
Residuals Anal	ysis						
Mean	0.0000		-6.026e-07		1.647e-0	8	
Std. dev	1.0000		0.0024		0.9786		
J-Bera	124.3 (0.00	00)	0.0128 (0.99	36)	4.8301 (0.0894)	
QB(12)	27.00 (0.00	00)	19.448 (0.07	'83)	11.306 (0.5028)		
QB2(12)	15.496 (0.2	154)	17.319 (0.13	(79)	13.955 (0.3036)	
ARCH(12)	3.4014 (0.0002)		2.8625 (0.00	14)	1.3775 (0.1824)		

specification is also confirmed by the RCM criterion, which is the closest to zero, e.g., 1.840. For instance, the RCM criterion indicates small values for both the MS(2) - AR(1) and MS(3) - AR(1) specifications, compared to the value of 100 when the true DGP is a linear model. Moreover, by comparing the RCM of the two specifications, the MS(3) - AR(1) and MS(2) - AR(1) models, the results show that the MS(3) - AR(1) model is more appropriate and fits the evolution of the tourist arrival time series better; e.g., the RCM of the MS(3) - AR(1) model is equal to 1.840, which is lower than the value of 8.355 for the MS(2) - AR(1) model.

In addition to the previous LR tests and RCM criteria, we compared these three models based on their residuals properties. The results of the residual analysis, reported at the bottom of Table 4, show that MS(3) - AR(1) is the only specification where the hypothesis of the absence of autocorrelation and homoscedasticity cannot be rejected.

To summarize, the different tests and techniques used to select the model that best fits the tourist arrival time series support with strong evidence a Markov switching model with three states in the intercept, trend, and variance.

5.2.2. Interpretation of the 3S-MS-AR(1) model

Turning now to the interpretation of the results of the 3S-MS-AR(1) selected model, the results show that all the estimated coefficients (μ_1 ,

(footnote continued)

 μ_2 , μ_3 , φ_1 , δ_1 , δ_2 , δ_3 , σ_1^2 , σ_2^2 , σ_3^2 , and p_{ii}) for i = 1, 2, 3 are highly significant at the 1% level of significance.

The results reported in Table 4 can be interpreted as follows:

- 1. Regime 1 corresponds to the period where Tunisian tourist activity is characterized by a period of tourism activity expansion and where the number of tourist arrivals is at its highest level. For instance, regime 1 has the highest value of the estimated intercept coefficient, $\mu_1 = 10.737$, the lowest value of volatility, $\sigma_1^2 = 0.041$, and an estimated coefficient for the upward trend that it is not well pronounced, $\delta_1 = 0.0022$. Economically, this regime corresponds to a period of economic expansion and periods where the local political environment is stable.
- 2. Regime 2 corresponds to the period where Tunisian tourism activity is—on average—high: the estimated intercept coefficient is $\mu_2 = 10.628$. However, the period of regime 2 is characterized by a high level of volatility, $\sigma_2^2 = 0.0714$, compared to the first regime. This means that this regime corresponds to an intermediate regime characterized by an average level of tourist arrivals that is similar to that of regime 1 and by a level of volatility that is similar to that of regime 3. In terms of trend evolution, this regime has the lowest trend coefficient of $\delta_2 = 0.0017$. This regime corresponds to the periods after September 11th, 2001; after the Jasmine revolution and after the recent terrorist attacks of 2016.
- 3. The results show that regime 3 corresponds to the period in which the Tunisian tourism activity is in its lower level (period of recession). This regime is characterized by the lowest estimated value of the intercept coefficient, $\mu_3 = 10.118$, and the highest value of the variance, $\sigma_3^2 = 0.0762$, compared to the two previous regimes. However, this regime is characterized by the highest estimated trend coefficient, i.e., $\delta_3 = 0.0030$. Economically, this regime corresponds to the period in which the tourism sector was subject to important negative shocks, such as the period of the Jasmine revolution and the period of the two terrorist attacks of 2015.

For all three regimes, the type and exact dates of these shocks will be discussed and analyzed in the next section.

Now, once the specification MS(3) - AR(1) has been approved as the most accurate and appropriate specification to describe the evolution of the tourist arrivals time series, we will focus our analysis on the economic interpretations and the economic and management implications of these results.

This will be discussed first on the basis of the regime classification and then on relaying the dates of the breaks of the major political shocks and terrorist attacks that have affected Tunisia.

5.3. Probabilities smoothing and regime classification analysis

As discussed previously, the smoothed probability for each regime is the tool used to date the breaks (see Hamilton, 1989). The smoothed probabilities for the selected MS(3) - AR(1) model are reported in Fig. 3. The first box, in the top left, reports the time series of actual seasonally adjusted monthly tourist arrivals in red and the three regime periods in the shaded area. These three shaded areas are also presented in the three other boxes with a box for each regime duration. From Fig. 3, we can conclude that the number of tourist arrivals in the first regime is more stable in terms of duration and variability compared to the two other regimes, followed by regime 2 and lastly regime 3. Moreover, Fig. 3 shows clear evidence that regimes 2 and 3 are more volatile than regime 1 and that regime 3 is characterized by the lowest level of the number of tourist arrivals.

In addition, the results of the regime classification reported in Table 5 show that the first regime spans exactly 8 years and 10 months (106 months), divided into two sub-periods where the first sub-period spans 21 months and the second spans 85 months. The second regime spans 73 months, divided into three sub-periods of 26 months for the

upper bound test to determine the number of regimes. Hansen (1992) used empirical process theory to bound the asymptotic distribution of the standardized Likelihood Ratio statistic. Then he used the simulation to determine the critical values for some cases of a Markov switching process based on the grid search technique. Garcia (1998) tabulated some critical values for the 2-State-MS-AR models under some particular assumptions.



Fig. 3. Monthly number of tourists arrivals and probabilities smoothing from the MS(3) - AR(1)

Table 5	
Regime classification and duration	based on smoothed probabilities.

Regime	Starting date	Ending date	Duration
Regime 1	2000-M01	2001-M09	21 Months (1 year and 9 month)
	2003-M12	2010-M12	85 Months (7 years and 1 month)
Regime 2	2001-M10	2003-M11	26 Months (2 years and 2 months)
	2011-M09	2015-M04	44 months (3 years and 8 months)
	2016-M07	2016-M09	3 months
Regime 3	2011-M01	2011-M08	8 months
	2015-M05	2016-M06	14 months (1 year and 2 month)

first sub-period, 44 months for the second sub-period and 3 months for the last sub-period. Finally, the last regime spans 22 months, divided into two sub-periods of 8 months for the first sub-period and 14 months for the second sub-period.

5.4. Breaks dates - political and/or terrorist attacks

The results of the break dates and their corresponding potential events are reported below in Table 6. In particular, five breaks have been identified corresponding to October 2001, December 2003, January 2011, September 2011, May 2015, and June 2016. As shown in Table 6, all five determined breaks correspond to major political and/or terrorists shocks that have negatively affected the Tunisian tourism activity.

These shocks are detailed as follows:

Table 6

Breaks	dates	and	maior	political	events	and/or	terrorist	attacks.
Dicuno	uuico	unu	major	ponticui	C V CIIILO	una, or	terroriot	actuents.

Break date	Political events and/or terrorist attacks
October 2001	September 2001 terrorist attacks in NewYork.
December 2003	End of the 09/11 and Djerba 05/2002 terrorist attack.
January 2011	Jasmine revolution, January 2011.
September 2011	The end of the effect of Jasmine revolution, January 2011 shock.
May 2015	Bardo National Museum and Sousse Attacks of March and May 2015.
June 2016	The end of the effect of the Bardo National Museum and Sousse Attacks

- 1. The first two break dates, October 2001 and December 2003, are associated with the beginning and ending of the first sub-period of regime 2. The first break of October 2001 corresponds to the September 11 attack (2001). This international shock seems to have had a significant impact on the number of tourist arrivals in Tunisia. However, as this break date fall into the intermediate regime (regime 2), then one can conclude that the effects of international shocks on the Tunisian tourism activity are less severe compared to the local shocks (political and terrorists attacks) that characterize regime 3. The duration of the October 2001 break was extended in terms of duration by the Djerba terrorist attack of May 2002. The second break of December 2003 is just the break date corresponding to the end of the October 2001 shock. In terms of duration, the effects of this break persisted for 26 months.
- 2. The third break date corresponds to the Tunisian Jasmine revolution of January 2011. This shock seems to have the most severe effects on the Tunisian tourism activity compared to all the other shocks (see Fig. 3). This shock has the lowest duration, lasting for only 8 months. This can be explained by the fact that this shock happened in the low season of tourism activity. Moreover, this shock seems to have had a negative short effect due to the absence of security; however, it had a positive impact in the medium-long run, mainly once the security in the country was restored. This can be explained by the fact that tourists are generally interested in visiting countries with special "achievements", such as countries where people are revolting in favor of democracy. The break of September 2011 corresponds to the end of the impact of the Jasmine revolution shock on the tourism activity.
- 3. Finally, the last two break dates of May 2015 and June 2016 correspond to the Bardo National Museum terrorist attack on March 18, 2015, which was followed 3 months later on June 26, 2015 by the terrorist attack against two hotels at Port El Kantaoui, Sousse, a tourist resort. The duration of this sub-period of regime 3 is 14 months (1 year and 2 months). The largest effects in terms of shock duration, compared to the Jasmine revolution shock, can be explained by the difference in nature of these two breaks. For instance, whereas the Jasmine revolution can be considered as a political shock, the March–June 2015 shocks are considered terrorist attack shocks. Moreover, this difference in terms of duration can be explained by the increased number of terrorist attacks during 2015. These attacks can be divided into two groups: (1) attacks against a

touristic location, which directly affect tourist arrivals, such as the two shocks considered and discussed previously (Bardo and Sousse attacks), and (2) attacks against the Tunisian army and security forces, which do not directly affect the number of tourist arrivals.

5.5. Results robustness checking

This section is devoted to assessing the robustness of the empirical results to (1) the method of deseasonalization and (2) the proxy of tourism activity employed. As a second method of deseasonalization, we propose the use of the \times 12-ARIMA technique and as a second proxy of tourism activity, we propose the use of the number of overnights stays.

5.5.1. Robustness to deseasonalization method

As mentioned in subsection 2.2., the ×12-ARIMA technique will be used to examine the robustness of our results to different methods of deseasonalization. The results of using ×12-ARIMA as an alternative method of deseasonalization are reported in the supplementary document. Overall, the results regarding the unit root tests, long memory tests, and Shimotsu (2006a) tests are similar to those obtained when the TRAMO/SEATS technique is used to deseasonalize the tourist arrival series. The specification of the selected Markov switching model is also similar to that of the adjusted number of the tourist arrival series via the TRAMO/SEATS technique. The only important difference is that ×12-ARIMA failed to detect the break of September 2001. This can be explained by the fact that the TRAMO/SEATS technique performs better for a large sample size compared to ×12-ARIMA.

However, despite these differences, the results regarding the postrevolution shocks and their duration are the same as those detected using the TRAMO/SEATS technique.

5.5.2. Robustness to second proxy of tourism activity

The robustness of the empirical result is also examined by using the number of overnight stays as a second proxy of tourism activity. An important difference compared to the tourist arrival series is that this new proxy not only accounts for international overnight stays but also takes into consideration the overnight stays by local residents.

The same results as in sections 3 and 4 above are reported for the number of overnights stays in the supplementary document and for the two methods of deseasonalization. The results of the unit root tests with structural breaks do not show any significant differences compared to the results discussed above for the number of tourist arrivals. The results show evidence of stationarity for the monthly number of hotel rooms reserved when using unit root tests accounting for structural breaks. The date of the breaks are, overall, similar to those obtained in the case of the number of arrivals. Regarding the results of testing for long memory behavior, there is strong evidence for long memory behavior. However, when testing for true versus spurious long memory, the results show that the null hypothesis of true long memory is rejected when splitting the entire sample into 5 or 6 subsamples for the two values of the truncation parameter: m = 20 and m = 40. In line with the results of the unit root tests with structural breaks, which show evidence of stationarity with five breaks, the results of the splitting sample test of Shimotsu (2006a) confirm the evidence of short memory with structural breaks.

Regarding the results of the estimation of the Markov switching autoregressive model, the results show that the number of overnight stays is characterized by three-state Markov switching in both the intercept, trend, and variance coefficients. In addition, the results related to the dates of breaks confirm those obtained for the case of the monthly number of tourist arrivals, especially for the post-revolution period. However, the period before the Jasmine revolution is slightly different from that obtained for the number of tourist arrivals, especially in terms of the number of breaks, which is higher when using the TRAMO/SEATS method compared to the \times 12-ARIMA technique.

6. Conclusion

This paper examines the effect of terrorist attacks and political instability on tourism activity in Tunisia. In particular, it investigates the magnitude of the effects of the Jasmine revolution and recent terrorist attacks on the number of tourist arrivals and number of overnights stays. The preliminary results show that the data seem to be characterized by four statistical properties: seasonality, unit root, structural breaks, and long memory. As a first step, to examine these statistical properties in detail, we use the unit root tests with structural breaks of Charfeddine and Guégan (2011), the modified R/S and V/S non-parametric tests of long memory, and the estimated values of the fractional long memory parameter d obtained by using the GPH, GSP, LW, ELW, and 2SFELW estimation methods. We test for true versus spurious long memory behavior by using the splitting test of Shimotsu (2006a). In the second step, once the evidence for changing in regimes is validated, we estimate a Markov switching model with a Markov change in the intercepts, trends, and variances for the number of tourist arrivals and the number of nights stayed. The results of the selected model are used to date exactly the breaks and to inspect the effects of the tourist attacks and political instability on tourism activity.

The main contributions of this study can be summarized in the following three points. First, whereas previous studies focused on the two possible extreme cases of transitory versus permanent impacts of terrorists and political violence on tourism activity, this study adds to the literature on the persistence of shocks on tourism demand by examining the intermediate case where political and terrorist attacks have long-lasting effects. Second, this study allows us to differentiate empirically the effects of terrorist shocks and political violence shocks in terms of magnitude and duration. We find evidence of four significant shocks that occurred after the end of the year 2010, compared to only one break before that date. Our results reveal that the effect of terrorist attack shocks on tourism activity is more severe than that of the Jasmine revolution. We also detected that, in terms of time duration, terrorist attack shocks have a more lasting effect (1 year and 2 months) compared to the Jasmine revolution shock (8 months). Third, we distinguish between internal and external shocks. Our results suggest that, for the case of Tunisia, internal shocks have the highest impact on tourism activity, whereas external shocks, including the terrorist attacks of September 11 and the global financial crisis of 2008, which fall into the second regime, have only a moderate impact. This latter result can be explained by the type and nature of the tourism sector in Tunisia, which is known as a low-cost destination.

Finally, this study proposes and discusses several marketing, economic, and security policies that can be implemented by Tunisian policymakers to better respond to all shocks that can affect the tourism industry in Tunisia.

7. Policy implications

Our empirical findings have several important policy implications in how to attenuate and reduce the negative impacts of political and terrorist attacks on Tunisian tourism activity. The results obtained from the determination of the true DGP of the number of tourist arrivals and the estimation of the appropriate Markov switching model suggest that the Tunisian tourism decision-makers from the government and private tourism managers should choose with caution the timing of their response depending upon the shocks' occurrence and the estimated impact. Indeed, determining factors affecting tourism demand, such as political instability and terrorist upheavals, and estimating their time path impact are very helpful for policy decision-makers when drawing and designing effective strategies to restore the tourism activity. Consequently, the Tunisian government should put more importance on national security and address terrorist attacks and the media sector with more professionalism.

7.1. Security policy

Our empirical findings show that terrorist attacks have a long duration compared to other types of shocks. This is mainly because many countries, including Great Britain, Belgium, the Netherlands, Sweden, and Denmark, prohibited travelling to Tunisia after the terrorist attacks of June 2015. Therefore, we believe that Tunisian authorities must work closely with the diplomatic missions of the European countries in the field of intelligence and the fight against terrorism. In addition, Tunisian diplomatic missions abroad must have a policy of economic communication of the image of Tunisia to different host countries to reassure foreign partners in matters of security. We believe that the improvement of inside security and the level of cooperation and communication with foreign governments may help to attenuate the impact of shocks on tourism activity.

7.2. Marketing policy

Generally, the media tends to exaggerate the virulence of terrorist events, negatively altering the destination's image, which can engender a decrease in tourist arrivals (Liu & Pratt, 2017). In terms of policy implications, decision makers in the Tunisian tourism industry need to reassure foreign tour operators and promote marketing initiatives and media policy to recover the positive image of the country because tourists may feel various levels of risk aversion to terrorist events.

The marketing policy must cover both the short and long terms. In the short term, the national tourism office in cooperation with tour operators must develop an anti-crisis communication policy to improve the image of Tunisia as an excellent tourism destination. Some international communication agencies established in the host countries must be involved in foreign media to temper the ardor of terrorist attacks and preserve the image of Tunisia as a safe country. In the long term, tourist operators in Tunisia must develop a commercial strategy aimed at extending the markets and reducing the risks of an over-dependence of tourist arrivals from the traditional Western markets by developing new opportunities from Eastern Europe and Asia (Czech Republic, Russia, Ukraine, China, etc.). Another possibility would be to promote Muslimoriented tourism and develop in some regions such as Djerba, for example, new products closer to Muslim religious sensibilities.

7.3. Monetary policy

Tourists are generally risk averse to terrorist events. In such

Appendix

1. Unit root tests with structural breaks of Carrion-i-Silvestre et al. (2009a).

The model considered by Carrion-i-Silvestre et al. (2009a) is the following,

$$y_t = d_t + \varepsilon_t$$
(A1)
$$\varepsilon_t = \alpha \varepsilon_t + \nu_t \quad t = 0, ..., T$$
(A2)

where y_i is a time series, $\{\varepsilon_t\}$ is the error term with mean zero. ε_0 is supposed to be equal to 0. The disturbance v_t is defined as $v_t = \sum_{i=0}^{\infty} \eta_i \eta_{t-i}$, with $\sum_{i=0}^{\infty} i |\eta_i| < \infty$ and $\{\eta_t\}$ as a martingale difference sequence adapted to the filtration $F_t = \sigma - field\{\eta_{t-i}; i \ge 0\}$. The long and short run variances are defined as $\sigma^2 = \sigma_{\eta}^2 \gamma(1)^2$ and $\sigma_{\eta}^2 = \lim_{t \to \infty} T^{-1} \sum_{t=1}^{T} E(\eta_t^2)$, respectively.

The component d_t in (1) is given by the following,

 $d_t = z_t'(\lambda)\psi,$

where,

$$z'_t(\lambda) = [z'_t(T_0), z'_t(T_1), ..., z'_t(T_m)] \text{ and } \psi = (\psi'_0, \psi'_1, ..., \psi'_m)'$$

The deterministic component $z'_t(T_j)$ and the coefficient ψ define three principal models largely explored and used in the empirical literature. These models correspond to the crash (level shift), the slope change and the mixed change models.

In this paper, we focused our analysis in the crash and mixed change models (see Perron, 1989).⁴

⁴ See also Zivot and Andrews (1992), Ng and Perron (1995) and Lee and Strazicich (2003).

incidents, the Central Bank should adopt a more flexible exchange rate regime in the short term to reduce such risk aversion. The appreciation of foreign currencies vis-à-vis the Tunisian dinar should encourage some tourists less affected by security menaces to pursue their visit to affected destinations. This cyclical flexibility could reduce the trade deficit. Note that the Algerian market, which remains very faithful to Tunisia as a destination, generated nearly 1.5 million tourist admissions in 2015. Thanks to their extra hotel spending, they have played an important role in regional development dynamics, becoming the first tourist clientele in Tunisia. A Maghreb monetary union must start with Tunisia's Algerian neighbors, would could boost Algerian tourist arrivals in Tunisia. Moreover, Tunisia must reduce its monetary dependency on the strength of the euro by developing a partnership with the Chinese market and adopting its currency as a foreign currency to diversify Tunisia's trade policy.

7.4. Budgetary policy

There is also a need to implement a countercyclical budgetary policy that takes the volatility of foreign exchange earnings coming from the tourism industry into account following internal shocks and sets up an allocation process for recovery campaign funding. The creation of a fund to support tourism services would make it possible to reduce the negative impact of a terrorist event on tourism activity following the transitory decline of arrivals of tourists in Tunisia and its social consequences. This fund should be financed during the period of rising tourism receipts from exceptional tourist seasons to act as a "shock absorber" during transitional periods of crisis following the advent of political instability or terrorist attacks because the recovery is likely to be slow afterwards. This fund can have two goals. The first is a social policy aiming to reduce the severity of the temporary job losses in the tourism industry after terrorist events. The second is serving as an advertising operation, updating overseas communication policy according to markets, to restore the value of the Tunisian offer and the image of the destination.

Despite the important results found in this study and their policy implications discussed in the next section, the current study has some limitations that can be explored in future research. These limits include, for example, the short period of study in terms of the number of observations and the non-exploration of the performance of the models examined in terms of out-of-sample forecasting and in determining the true nature of the observed long memory behavior. As suggested above, these points can be explored in future research.

(A4)

(A3)

(A5)

(A6)

(A7)

(A8)

For the level shift model known as the crash model, the deterministic component $z'_t(T_j)$

$$z_t'(T_j) = DU_t(T_j)$$

for the mixed change model,

$$z_t'(T_i) = (DU_t(T_i), DT_t^*(T_i))'$$

For j = 0, $z'_t(T_0) \equiv z_t(0) = (1, t)'$ and $\psi_0 = (\mu_0, \beta_0)'$. For $1 \le j \le m$, $\psi_j = \mu_j$ in the crash model and $\psi_j = (\mu_j, \beta_j)'$ in the mixed change model. $DU_t(T_j) = 1$ and $DT_t^*(T_j) = (t - T_j)$ for $t > T_j$ and 0 otherwise. $T_j = [T\lambda_j]$, which denotes the *j*-th break date, with [.] defining the integer part, and $\lambda_j \equiv T_j/T \in (0,1)$ as the break fraction parameter. The authors use the sum of squared residuals (SSR) of the GLS-detrended model to estimate the break dates, the authors used the global minimization of the.

The four tests that we propose to use in this study are the counterpart tests of the Eliot et al. (1996) for the case of unit root tests without breaks. The tests are the $MZ_{\alpha}^{GLS}(\lambda)$, $MSB^{GLS}(\lambda)$, $MZ_{\tau}^{GLS}(\lambda)$ and $MP_{T}^{GLS}(\lambda)$ tests where λ is as defined above.

The limit distribution and the critical values of all these tests are obtained by simulation, see Carrion-i-Silvestre et al. (2009a) paper for more details.

2. Shimotsu (2006a) test of true versus long range dependence

The Shimotsu (2006a) test consists of splitting the tourist arrivals time series into b subsamples and in comparing the estimated long memory parameter d of the entire sample to the estimated fractional long memory parameter of each subsample. The idea of this test is based on the following assumption: If the observed long memory behavior is a true behavior, then the estimated fractional long memory parameter of *d* for the full sample should not be statistically different from those estimated from the subsamples (see Charfeddine, 2016; Abuzayed et al., 2018; and; Charfeddine & Maouchi, 2018).

The null hypothesis of true long memory I(d) is,

$$H_0$$
:True long memory $(H_0: d = d^{(1)} = ... = d^{(b)})$

The alternative hypothesis corresponds to,

H1:Short memory affected by breaks

where $d^{(i)}$ for $i \in \{1, 2, ..., b\}$ is the true value of the fractional long memory parameter from the ith subsample and d is the true value for the full sample.

The test statistic proposed by Shimotsu (2006a) is given by,

$$W_c = 4m \left(\frac{c_{m/b}}{m/b}\right) A \hat{d}_b \left(A\Omega A'\right)^{-1} \left(A\hat{d}_b\right)'$$
(A9)

where,

$$\hat{d}_b = \begin{pmatrix} \hat{d} - d \\ \hat{d}^{(1)} - d \\ \vdots \\ \hat{d}^{(b)} - d \end{pmatrix}, \quad A = \begin{pmatrix} 1 & -1 & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 1 & 0 & \cdots & -1 \end{pmatrix} \quad \Omega = \begin{pmatrix} 1 & \tau_b' \\ \tau_b & bI_b \end{pmatrix},$$

where I_b is a $(b \times b)$ identity matrix and τ_b is a $(b \times 1)$ vector of ones. *m* is the number of the periodogram ordinates used in the objective function. c_m is a number given by,

$$c_m = \sum_{j=1}^m v_j^2, \quad v_j = \log \lambda_j - \frac{1}{m} \sum_{j=1}^m \log \lambda_j = \log j - \frac{1}{m} \sum_{j=1}^m \log j$$

Shimotsu (2006a) shows that the limiting distribution of the splitting test W_c is the chi-squared limiting distribution with b - 1 degrees of freedom.

Note: The critical values for the W_c statistic for the Shimotsu (2006) is $\chi^2(b-1)$ where b is the number of subsamples. and $\chi^2(4) = 9.49$ and $\chi^2(5) = 11.070$ at the 5% level of significance and equal to $\chi^2(4) = 13.277$ and $\chi^2(5) = 15.086$ at the 5% level of significance.

Author contribution

Dr. Lanouar Charfeddine contribution to this paper includes writing the abstract and performing all the two sections of "Data and preliminary analysis" and "a Markov Switching model for the number of tourist arrivals". The contribution includes also the participation in writing the introduction, literature review, the tourism sector in Tunisia and conclusion sections as well as the discussion and validation of the policy implication section.

Prof. Mohamed Goaied contribution to this paper includes participation in writing the introduction, literature review, the tourism sector in Tunisia, and conclusion sections as well as the data collection and all the section of policy implication. The contribution includes also the discussion and validation of the empirical analysis.

416

References

Economics, 50(23), 2569-2599.

 Agiomirgianakis, G., Serenis, D., & Tsounis, T. (2017). Effective timing of tourism policy: The case of Singapore. *Economic Modelling*, 60, 29–38.
 Arana, J. E., & Leon, C. J. (2008). The impact of terrorism on the tourism demand. *Annals*

Abuzayed, B., Al-Fayoumi, N., & Charfeddine, L. (2018). Long range dependence in an emerging stock market's sectors: Volatility modelling and var forecasting. Applied

C. Lanouar, M. Goaied

of Tourism Research, 35(2), 299–315.

- Avraham, E. (2016). Destination marketing and image repair during tourism crises: The case of Egypt. Journal of Hospitality and Tourism Management, 28, 41–48.
- Baele, L. (2005). Volatility spillover effects in european equity markets. Journal of Financial and Quantitative Analysis, 40(02), 373–401.
- Bai, J., & Perron, P. (1998). Estimating and testing linear models with multiple structural changes. *Econometrica*, 66, 47–78.
- Bai, J., & Perron, P. (2003). Computation and analysis of multiple structural change models. Journal of Applied Econometrics, 18(1), 1–22.
- Barros, C. P., Gil-Alana, L. A., & Wanke, P. (2016). Energy Production in Brazil: Empirical facts based on persistence, seasonality and Breaks. *Energy Economics*, 54, 88–95.
- Ben Aissa, S., & Goaied, M. (2016). Determinants of Tunisian hotel profitability: The role of managerial efficiency. *Tourism Management*, 52, 458–478.
- Beran, J. (1994). Statistics for long-memory processes. Chapman and Hall/CRC.
- Carboni, M., Perelli, C., & Sistu, G. (2014). Is Islamic tourism a viable option for Tunisian tourism? Insights from Djerba. *Tourism Management Perspectives*, 11, 1–9.
- Carrasco, M. (2002). Misspecified structural change, threshold, and Markov-switching models. Journal of Econometrics, 109, 239–273.
- Carrion-i-Silvestre, J. L., Kim, D., & Perron, P. (2009). GLS-based unit root tests with multiple structural breaks both under the null and the alternative hypotheses. *Economic Theory*, 25, 1754–1792.
- Causevic, S., & Lynch, P. (2013). Political (in) stability and its influence on tourism development. *Tourism Management*, 34, 145–157.
- Charfeddine, L. (2016). Breaks or long range dependence in the futures energy volatility : Out-of-sample forecasting and VaR analysis. *Economic Modelling*, *53*, 354–374.
- Charfeddine, L., & Guégan, D. (2008). Is it possible to discriminate between different switching regressions models? An empirical investigation. *Euro-Mediterranean Economics and Finance Review*, 3(4), 54–75.
- Charfeddine, L., & Guégan, D. (2011). Which is the best for the US inflation time series: A structural change model or a long memory process? *IUP Journal Of Applied Economics*, 10(1), 5–25.
- Charfeddine, L., & Guégan, D. (2012). Breaks or long memory behaviour : An empirical investigation. *Physica A : Statistical Mechanics and Its Applications, 391*(22), 5712–5726.
- Charfeddine, L., & Maouchi, Y. (2018). Are shocks on the returns and volatility of cryptocurrencies really persistent? forthcoming: Finance research letter. https://doi.org/10. 1016/j.frl.2018.06.017.
- Chu, F. L. (2008). A fractionally integrated autoregressive moving average approach to forecasting tourism demand. *Tourism Management*, 29(1), 79–88.
- Cunado, J., Gil-Alana, L. A., & de Gracia, F. P. E. (2008 a). Fractional integration and structural breaks: Evidence from international monthly arrivals in the USA. *Tourism Economics*, 14(1), 13–23.
- Cunado, J., Gil-Alana, L. A., & de Gracia, F. P. (2008 b). Persistence in international monthly arrivals in the Canary Islands. *Tourism Economics*, 14(1), 123–129.
- Davies, R. B. (1977). Hypothesis testing when a nuisance parameter in present only under the alternative. *Biometrika*, 64, 247–254.
- Davies, R. B. (1987). Hypothesis testing when a nuisance parameter in present only under the alternative. *Biometrika*, 74, 33–43.
- Diebold, F. X., & Inoue, A. (2001). Long memory and regime switching. Journal of Econometrics, 105 131–159.
- Elliot, G., Rothenberg, J. T., & Stock, J. (1996). Efficient tests for an autoregressive unit root. *Econometrica*, 64(4), 813–836.
- Engel, R. F., & Smith, A. D. (1999). Stochastic permanent breaks. The Review of Economics and Statistics, 81, 553–574.
- Estrada, M. A. R., & Koutronas, E. (2016). Terrorist attack assessment: Paris november 2015 and Brussels March 2016. *Journal of Policy Modeling*, *38*(3), 553–571.
- Franses, P. H., Paap, R., & Fok, D. (2005). Performance of seasonal adjustment procedures: Simulation and empirical results. Econometric institute report, EI.
- Garcia, R. (1998). Asymptotic null distribution of the likelihood ratio test in Markov switching models. *International Economic Review*, 39, 763–788.
- Gasmi, A., & Sassi, S. (2015). International tourism demand in Tunisia: Evidence from dynamic panel data model. *Economics Bulletin*, 35(1), 507–518.
- Geweke, J., & Porter-Hudak, S. (1983). The estimation and application of long memory time series models. *Journal of Time Series Analysis*, 4, 221–238.
- Gil-Alana, L. A. (2005). Modelling international monthly arrivals using seasonal univariate long-memory processes. *Tourism Management*, 26(6), 867–878.
- Gil-Alana, L. A., De Gracia, F. P., & CuÑado, J. (2004). Seasonal fractional integration in the Spanish tourism quarterly time series. *Journal of Travel Research*, 42(4), 408–414.
- Gil-Alana, L. A., Mervar, A., & Payne, J. E. (2015). Measuring persistence in Croatian tourism: Evidence from the adriatic region. *Applied Economics*, 47(46), 4901–4917.
- Gil-Alana, L. A., Mervar, A., & Payne, J. E. (2016). Modeling the degree of persistence in Croatian tourism. *Tourism Economics*, 22(3), 655–664.
- Gil-Alana, L. A., Mudida, R., & de Gracia, F. P. (2014). Persistence, long memory and seasonality in Kenyan tourism series. Annals of Tourism Research, 46, 89–101.
- Gil-Alana, L. A., Cunado, J., & Perez de Gracia, F. (2008). Tourism in the canary islands: Forecasting using several seasonal time series models. *Journal of Forecasting*, 27(7), 621–636.
- Giraitis, L., Kokoszka, P., Leipusc, R., & Teyssiere, G. (2003). Rescaled variance and related tests for long memory in volatility and levels. *Journal of Econometrics*, 112(2003), 265–294.
- Gourieroux, C., & Jasiak, J. (2001). Memory and infrequent breaks. *Economics Letters*, 70(2001), 29–41.
- Granger, C. W. J., & Hyung, N. (2004). Occasional structural breaks and long memory with application to the S&P500 absolute stock returns. *Journal of Empirical Finance*, 11, 399–421.
- Granger, W. J., & Terusvirta, T. (1999). A simple nonlinear time series model with

misleading linear properties. Economics Letters, 62, 161-165.

- Hamilton, J. D. (1989). A new approach to the economic analysis of nonstationary time series and the business cycle. *Econometrica*, 57, 357–384.
- Hamilton, J. D. (1990). Analysis of time series subject to change in regime. Journal of Econometrics, 45, 357–384.
- Hansen, B. E. (1992). Tests for parameter instability in regression I1 processes. Journal of Business and Economic Statatistics, 10, 321–335.
- Hood, C. C., Ashley, J. D., & Findley, D. F. (2000). An empirical evaluation of the performance of TRAMO/SEATS on Simulated Series. American statistical association proceedings of the business economics section.
- Hylleberg, S., Engle, R. F., Granger, C. W., & Yoo, B. S. (1990). Seasonal integration and cointegration. *Journal of Econometrics*, 44(1), 215–238.
- IMF (2017). Turkeys-economy-hit-by-declining-tourism. February: IMF News. https://www. imf.org/en/News/Articles/2017/02/21/NA230217.
- Jlassi, K., Fliti, Z., & Chaibi, H. (2015). The Tunisian tourism business : What we learn about the european demand ? Journal of Applied Business Research, 31(6), 2079–2090.
- Krolzig, H. M. (1997). Markov switching vector autoregressions : Modelling, statistical inference and application to business cycle analysis, lecture Notes on Economics and Mathematical Systems, Vol. 454. Berlin: Springer.
- Lean, H. H., & Smyth, R. (2009). Asian financial crisis, avion flu and terrorist threats : Are shock to malaysian tourist arrivals permanent or transitory? Asia Pacific Journal of Tourism Research, 14(3), 301–321.
- Lee, J., & Strazicich, M. C. (2003). Minimum Lagrange multiplier unit root test with two structural breaks. The Review of Economics and Statistics, 85, 1082–1089.
- Liu, A., & Pratt, P. (2017). Tourism's vulnerability and resilience to terrorism. *Tourism Management*, 60(2017), 404–417.
- Lo, A. W. (1991). Long-term memory in stock market prices. Econometrica, 59, 1279–1313.
- Lumsdaine, R. L., & Papell, D. H. (1997). Multiple trend breaks and the unit root hypothesis. The Review of Economics and Statistics, 79(1997), 212–218.
- Narayan, P. K. (2005). Testing the unit root hypothesis when the alternative is a trend break stationary process: An application to tourist arrivals in Fiji. *Tourism Economics*, 11(3), 351–364.
- Narayan, P. K. (2008). Examining the behaviour of visitor arrivals to Australia from 28 different countries. *Transportation Research Part A: Policy and Practice*, 42(5), 751–761.
- Neumayear, E. (2004). The impact of political violence on tourism. Journal of Conflict Resolution, 48(2), 259–281.
- Ng, S., & Perron, P. (1995). Unit root tests in ARMA models with data dependent methods for the selection of the truncation lag. *Journal of the American Statistical Association*, 90, 268–281.
- Ohanissian, A., Russell, J. R., & Tsay, R. S. (2008). True or spurious long memory? A new test. Journal of Business & Economic Statistics, 26(2), 161–175.
- Ouerfelli, C. (2008). Co-integration analysis of quarterly European tourism demand in Tunisia. *Tourism Management*, 29(1), 127–137.
- Perron, P. (1989). The great crash, the oil price shock and the unit root hypothesis. Economic Times, 57, 1361–1401.
- Peter, C., Poulston, J., & Losekoot, E. (2014). Terrorism, rugby, and hospitality: She' ll be right. Journal of Destination Marketing & Management, 3(4), 253-261.
- Quandt, R. (1958). The estimation of the parameter of a linear regression system obeying two separate regimes. *Journal of the American Statistical Association*, 53, 873–880.
- Raza, S. A., & Jawaid, S. T. (2013). Terrorism and tourism: A conjunction and ramification in Pakistan. *Economic Modelling*, 33, 65–70.
- Reisenger, Y., & Mavondo, F. (2005). Travel anxiety and intentions to travel internationally: Implications of travel risk perception. *Journal of Travel Research*, 43, 212–225
- Robinson, P. M. (1995). Gaussian semiparametric estimation of long range dependence. Annals of Statistics, 24, 1630–1661.

Robinson, P. M., & Henry, M. (1999). Long and short memory conditional heteroskedasticity in estimating the memory parameter of levels. *Econometric Theory*, 15(issue 03), 299–336.

- Shimotsu, K. (2006a). Simple (but effective) tests of long memory versus structural breaks. Department of Economics, Queen's University Working paper.
- Shimotsu, K. (2006b). Exact local White estimation of fractional integration with unknown mean and time trend. Queen's Economics Department working paper No. 1061. Available at: http://qed.econ.queensu.ca/pub/faculty/shimotsu/.
- Shimotsu, K., & Phillips, P. C. B. (2005). Exact local Whittle estimation of fractional integration. Annals of Statistics, 33, 1890–1933.
- Song, H., & Li, G. (2008). Tourism demand modelling and forecasting—a review of recent research. *Tourism Management*, 29(2), 203–220.
- Sonmez, S. F. (1998). Tourism, terrorism and political instability. Annals of Tourism Research, 25(2), 416–456.
- Terasvirta, T. (1994). Specification, estimation, and evaluation of smooth transition autoregressive models. Journal of the American Statistical Association, 89(425), 208–218.
- Terasvirta, T., & Yang, Y. (2014). CREATES Research Papers. Department of Economics and Business Economics, Aarhus University CREATES Research Papers 2014-08.
- Tong, H. (1990). Nonlinear time series : A dynamic system approach. Oxford University Press.
- Walther, T., Klein, T., Thu, H. P., & Piontek, K. (2017). True or spurious long memory in European non-EMU currencies. *Research in International Business and Finance*, 40, 217–230.
- Wolff, K., & Larsen, S. (2014). Can terrorism make us feel safer? Risk perceptions and worries before and after the July 22nd attacks. Annals of Tourism Research, 44, 200–209.
- Zivot, E., & Andrews, D. W. K. (1992). Further evidence on the great crash, the oil shock and the unit root hypothesis. Journal of Business & Economic Statistics, 10, 251–270.



Dr. Charfeddine Lanouar is associate Professor of economics and econometrics at Qatar University. He holds a Ph.D in applied Econometrics in 2007 from University Paris II – Pantheon - Assas, France. His research area include nonlinear models (Switching and long memory processes), financial modeling and forecasting, energy economics, and fiscal and monetary policies. He has published several articles in these topics in internationally respected peer-reviewed journals including the journal of Energy Economics, Energy, Energy Policy, Economic Modeling, Emerging Market Review and Cleaner Energy Production etc ... His research was funded by internal and external grants including the QFC and QNRF grants.



Prof. Mohamed Goaied is a professor of Economics and Statistics at Qatar University. He holds a Ph.D in Mathematical Economics and Econometrics from University of Dijon, France. He had a position as a full professor of Econometrics and statistics at IHEC Carthage from 2001 to 2016 and headed the laboratory of Research of Applied Economics and Finance, at University of Carthage, from 2005 to 2016. He has carried out research and consulting activities in the field of Industrial Economics, Labour Econometric Reviews, Journal of Hospitality & Tourism Research, Journal of Productivity Analysis.